

Since January 2020 Elsevier has created a COVID-19 resource centre with free information in English and Mandarin on the novel coronavirus COVID-19. The COVID-19 resource centre is hosted on Elsevier Connect, the company's public news and information website.

Elsevier hereby grants permission to make all its COVID-19-related research that is available on the COVID-19 resource centre - including this research content - immediately available in PubMed Central and other publicly funded repositories, such as the WHO COVID database with rights for unrestricted research re-use and analyses in any form or by any means with acknowledgement of the original source. These permissions are granted for free by Elsevier for as long as the COVID-19 resource centre remains active.



Contents lists available at ScienceDirect

Digestive and Liver Disease



journal homepage: www.elsevier.com/locate/dld

Liver, Pancreas and Biliary Tract

COVID-19 vaccination among cirrhotics in Italy: High coverage and effectiveness of 3 doses versus 2 in preventing breakthrough infection and hospitalization *



Tommaso Stroffolini^a, Alessia Ciancio^b, Alessandro Federico^c, Rosa G. Benigno^d, Guido Colloredo^e, Anna Lombardi^f, Grazia Anna Niro^g, Gabriella Verucchi^h, Luigina Ferrignoⁱ, Federico Gioli^j, Massimo Marignani^{j,*}

ARTICLE INFO

Article history: Received 30 September 2022 Accepted 21 November 2022 Available online 16 December 2022

Keywords: Cirrhosis Covid-19 Italy Vaccine

ABSTRACT

Background and Aims: Few reports, all retrospective, have evaluated vaccine coverage against COVID-19 infection in cirrhotic subjects. No data are available for European Countries. We aimed to explore this topic and potential independent predictors of lack of vaccination.

Methods: Between January 1st and June 30th 2022, 1512 cirrhotic subjects of any etiology were consecutively enrolled in an observational - prospective study in 8 referral centers in Italy. Adjusted Odds Ratios (O.R.) for the association with lack of vaccination and with occurrence of breakthrough infection were evaluated by multiple logistic regression analysis.

Results: Overall vaccine coverage was 89.7% (80% among people born abroad). Among the 1358 vaccinated people, 178 (13.1%) had a breakthrough infection; of them 12 (6.7%) were hospitalized, but none died. Independent predictors associated with lack of vaccination were birth abroad, age <65 years and lower years of schooling. Child stage B/C was the only independent predictor of breakthrough infection. Occurrence of breakthrough infection was more likely reported in subjects who received 2 doses of vaccine than in those who received 3 doses (33.9% versus 9.0%; *P*<0.001).

Conclusion: High vaccine coverage against COVID-19 infection is observed among cirrhotic subjects in Italy. Vaccine is effective in preventing severe outcomes. Three doses are more effective than two, even in cirrhotic subjects.

Lay Summary: This large cohort study evidenced high vaccine coverage against COVID-19 infection among cirrhotic subjects in a European country and the effectiveness of vaccine in preventing severe outcomes. Three doses of vaccine are more effective than two in preventing breakthrough infection and hospitalization. Informative campaigns targeting people younger than 65 years of age and those with lower years of schooling may increase these excellent results.

© 2022 Editrice Gastroenterologica Italiana S.r.l. Published by Elsevier Ltd. All rights reserved.

1. Introduction

E-mail address: mmarignani@hotmail.com (M. Marignani).

Subjects with liver cirrhosis, once infected with COVID-19 virus, have high rate of hospitalization, liver decompensation and death [1]. Among patients with cirrhosis, COVID-19 was associated with a 2.38-fold (95% C.I., 2.18–2.59) increased risk of a 30-day mortality,

1590-8658/© 2022 Editrice Gastroenterologica Italiana S.r.l. Published by Elsevier Ltd. All rights reserved.

^a Department of Tropical and Infectious Diseases, Policlinico Umberto I, Rome, Italy

^b Department of Gastroenterology, Ospedale Molinette, Torino, Italy

^c Hepato-Gastroenterology Unit, University of Campania Luigi Vanvitelli, Naples, Italy

^d Liver Unit, Department of Internal Medicine, ARNAS Garibaldi, Catania, Italy

^e Internal Medicine Unit, Policlinico S. Pietro, Bergamo, Italy

^fLiver Unit, Cardarelli Hospital, Naples, Italy

g Gastroenterology Unit, Fondazione Casa Sollievo della Sofferenza IRCCS, San Giovanni Rotondo, Italy

^h Department of Infectious Diseases, S. Orsola Hospital, Bologna, Italy

ⁱ National Health Institute, National Center for Global Health, Rome, Italy

^j Department of Digestive and Liver Disease, AOU S. Andrea, Rome, Italy

^c Collaborating group: Yulia Troshina^b, Marcello Dallio^c, Francesco Di Costanzo^f, Rosa Cotugno^g, Lorenzo Badia^h

 $^{^{\}ast}$ Corresponding author at: Department of Digestive and Liver Disease, AOU S. Andrea, Rome, Italy.

and among patients with COVID-19 and chronic liver disease, presence of cirrhosis was associated with a 3.31-fold (95% C.I., 2.91– 3.77) risk of 30-day mortality [2]. A key point of these severe outcomes is the innate and adaptive immune dysregulation present in subjects with liver cirrhosis [3]. Another important step is played by bacterial translocation from the intestinal lumen to mesenteric lymph nodes, able to increase the level of endotoxins and cytokines [4].

Development of effective vaccines against COVID-19 has been rapid. These have been highly effective in phase 3 clinical trials, offering a 94 to 95% protection from infection [5,6]. As for the other vaccines (such as those against *Streptococcus* pneumoniae [7], influenza [8] and hepatitis B virus [9], subjects with liver cirrhosis show a suboptimal antibodies response even against COVID-19 vaccine [10] as consequence of their immune dysfunction [3].

A cohort study of US veterans with cirrhosis showed that administration of 2 doses of mRNA vaccines was associated with a 100% reduction in hospitalization and death. However, this reduced schedule had a delayed and lower (78.6%) efficacy against COVID-19 breakthrough infection [11]. A more recent survey has confirmed the strong reduction of mortality in subjects with post vaccination COVID-19 infection, among both those with either compensated (HR, 0.19: 95% C.I. 0.08–0.45) and decompensated (HR, 0.27: 95% C.I. 0.08–0.90) cirrhosis [12]. Accordingly, international liver societies guidelines (i.e. AASLD and EASL) recommend COVID-19 vaccination for subjects with chronic liver disease [13,14].

Thus, knowledge of COVID-19 vaccine coverage and characteristics of subjects unvaccinated are of paramount importance to address and improve vaccination policy.

Few and retrospective reports have explored this topic. National data from the Veterans Health Administration in US has shown that 60% of cirrhotic patients received a SARS-COVID-2 vaccination [15]. In eastern China only 37.1% of decompensated cirrhotic subjects received at least one dose of vaccination, while 62.9% remained unvaccinated despite vaccination in this population has been shown to be safe [16].

Aim of this survey was to evaluate, for the first time in a European country, vaccine coverage and potential predictors associated with lack of vaccination against COVID-19 infection in a large sample of cirrhotic subjects recruited in 8 liver units across Italy. Attention has also been addressed to the rate and predictors of breakthrough infection among subjects with completion of the primary vaccination cycle.

2. Patients and methods

2.1. Study design and patients

This is a multicentre observational study, prospectively enrolling all consecutive patients with cirrhosis of any etiology. The study was performed in 8 tertiary centers (3 in the North, 1 in the centre, 3 in the South and 1 in Sicily) during a six-month period, from January 1st to June 30th, 2022. Seven liver units and 1 infectious disease unit participated. All cirrhotic subjects, regardless of the etiology, aged 18 years or older, consecutively observed as either inpatients or outpatients, were prospectively recruited. Incident cases were defined as those who were first diagnosed with liver cirrhosis during the study period; "prevalent cases" as those observed during the study period but with an already known diagnosis of liver cirrhosis. Subjects with hepatocellular carcinoma and/or liver transplantation were excluded from the study.

Subjects were classified as being vaccinated if they reported that they had received Coronavirus vaccine in the past. Patients were classified as having post-vaccination COVID-19 if the infection was diagnosed more than 14 days after the full vaccination. Ascertainment of COVID-19 infection was determined by a protease COVID-19 polymerase chain reaction (PCR) obtained by nasal swab.

Patients were evaluated only once at their first observation. In this occasion, a questionnaire was administered to subjects. Once evaluated, subjects were no longer followed-up.

Data were collected on an anonymous form at each participating center. The questionnaire contained information on sociodemographic characteristic of subjects, etiology and stage (Child score) of liver cirrhosis, referral pattern, self-reported history of vaccination, outcomes in vaccinated subjects, and potential reasons for vaccination or not vaccination against COVID-19.

All subjects gave their written consent to participate to the study, which was carried out in conformity with the 2013 revision of the Declaration of Helsinki. The study was approved by the ethic committee of Casa Sollievo della Sofferenza Hospital – IRCCS – San Giovanni Rotondo.

2.2. Diagnostic criteria

Liver cirrhosis was based on the presence of the peculiar clinical, biochemical, ultrasound signs [17] and transient elastometry performed by Fibroscan [18]. Etiology of liver cirrhosis was based on the criteria stated in a previous published survey [19].

2.3. Serological assays

Viral hepatitis serum markers (HBsAg and HCV) were determined by commercial immune-enzymatic assays (ELISA) in the laboratories of the eight hospital participating in the survey. The kits used were: HBsAg EIA Abbott for HBsAg detection and Ortho HCV 3.0 ELISA for HCV detection.

Clinical COVID-19 infection was laboratory-confirmed by a positive result of COVID-19 PCR by nasal swab.

2.4. Statistical analysis

Differences in proportions were evaluated by the Chi-squared test. A p value <0.05 was considered to be significant.

Crude Odds Ratio (OR) and their 95% Confidence Intervals (CI) for the association of outcome variable (vaccination status) with the socio-demographic and clinical characteristics of the subjects, were calculated by univariate analysis. In order to control for the disturbing influence of confounders, the independent predictors of lack of Coronavirus vaccination were identified by multiple logistic regression analysis. In the model, lack of vaccination was the outcome variable; age, area of birth, area of residence, years of schooling, etiology and Child stage were the forced variables. For the Odds ratios estimates, the reference category for each variable was that reporting the higher vaccination coverage rate.

A further logistic regression analysis was applied only to 3 doses vaccinated people to identify characteristics of subjects independently associated with breakthrough infection. In the model occurrence of breakthrough infection was the outcome variable, while sex, age, etiology and Child stage were the forced variables.

3. Results

3.1. Main characteristic of subjects

A total of 1512 cirrhotic subjects were recruited during the 6-months enrolment period. The male to female ratio was 2.3; mean age was 64.2 years, with most cases (49.7%) being older than 64 years of age. People born abroad were 10.3%. A low rate (9.1%) of subjects were incident cases. The majority were outpatients (80.0%). Nearly three quarter of cases (74.6%) had a viral etiology. Most subjects (83.7%) had compensated cirrhosis (Child A). (Table 1).

Table 1

Baseline characteristics of 1512 subjects with liver cirrhosis. Italy, 2022.

Characteristic		N.*	%
Sex	Male	1052	69.6
	Female	460	30.4
	Sex ratio (M/F)	2.3	
Age	Mean	64.2	
	Standard Deviation	13.1	
	Median	64	
	Range	21-97	
Age distribution	\leq 50 years	201	13.3
	50-64 years	559	37.0
	\geq 65 years	750	49.7
Area of birth	Italy	1356	89.7
	Abroad	155	10.3
Area of residence	North/Center	810	53.6
	South/Islands	702	46.4
YEARS OF SCHOOLING	≤8 years	703	46.7
	> 8 years	803	52.3
Type of case	Incident	137	9.1
	Prevalent	1374	90.9
Referral pattern	Inpatient	302	20.0
	Outpatient	1210	80.0
Etiology**	Viral	1129	74.6
	Alcoholic	253	16.8
	NAFLD/NASH	235	15.6
	Other	136	9.0
Child	A	1264	83.7
	В	190	12.6
	С	57	3.8

* For some variables, inconsistencies are due to missing values.

** Several cases had more than 1 etiological factor.

3.2. Vaccination status

In Italy, the third dose of mRNA vaccine was available for highrisk subjects since August 2021.

As many as 1358 (89.8%) subjects reported previous vaccination against COVID-19; 1137 (83.7%) of them received 3 or more doses of vaccine, 221 (16.3%) 2 doses, while none had received only one dose. Almost all (95.7%) had received RNA messenger vaccines. Among the 1358 people vaccinated, 178 (13.1%) had a breakthrough infection; 12 (6.7%) of them were hospitalised, but none died.

The mean time elapsed since the administration of the latest dose of vaccine and occurrence of breakthrough infection was 122 days (range 102–156). In the 12 subjects requiring hospitalization, the clinical presentation was respiratory distress in 9 cases, and liver decompensation in 3 cases (data not shown). Subjects who had received 2 doses of vaccine more likely reported breakthrough infection and hospitalization than those who had received 3 dose of vaccine (33.9% vs. 9.1%, P<0.001 and 4.6% vs. 0.5%, P<0.001, respectively).

Refusal of vaccination was reported by 133 (86.4%) of the 154 unvaccinated people; negative advice from the General Practitioner (G.P.s) against vaccination was reported by 10.4% of subjects. (Table 2)

Prevalence of vaccination coverage was lower in patients younger than 65 years of age (86.1% vs. 93.6%; p < 0.001), in those born abroad (80.0% vs. 90.9%; p < 0.001), in South Italy and Sicily residents (86.3% vs. 92.8%; p < 0.001), in subjects with lower years of schooling (86.6% vs. 92.8%; p < 0.001), among those with an alcoholic etiology (84.9% vs. 91.0%; p < 0.05), and in patients with Child B/C stages (83.8% vs. 91.0%; p < 0.05) (Table 3). Sex was the only significantly different characteristic among the 154 unvaccinated people by area of birth (54.8% in males and 45.2% in females; p < 0.05) (Table 4).

3.3. Independent predictors of lack of vaccination

The crude O.R.s for the association of the characteristics of subjects not undergoing COVID-19 vaccination, evidenced a link with all the variables considered (age, area of birth, area of residence, years of schooling, etiology and Child stage). After adjustment by multiple logistic analysis for the confounding effect of all variables considered, age <65 years (OR = 2.88; CI 95% = 1.91–4.36), being born abroad (OR = 2.31; CI 95% = 1.43–3.75), and lower years of schooling (OR = 1.91; CI 95% = 1.25–2.92) resulted independent predictors of lack of vaccination. Area of residence, etiology and Child stage were no longer associated (Table 5).

Table 2

Vaccination status against COVID-19 of 1512 subjects with liver cirrhosis. Italy, 2022.

Characteristic		N.	%
VACCINATED	Yes	1358	89.8
	No	154	10.2
Vaccinated*	< 3 doses	221	16.3
	\geq 3 doses	1137	83.7
Type of vaccine	Adeno/vector	42	3.1
	RNA messenger	1299	95.7
	Both	17	1.3
INFECTION IN VACCINATED	Yes**	178	13.1
	No	1180	86.9
INFECTION BY VACCINATION DOSE**	With < 3 doses	75	33.9
	With \geq 3 doses	103	9.1
HOSPITALIZATION IN VACCINATED	Yes	12	6.7
	No	166	93.3
HOSPITALIZATION BY VACCINATION	With < 3 doses	7	4.6
Dose ****./****	With \geq 3 doses	5	0.5
REASON OF VACCINATION	Advised by the G.P.	273	20.1
(N = 1358)	Advised by the specialist	701	51.6
	Self-reporting	384	28.3
REASON OF NOT VACCINATION	Lack of information	5	3.2
(N = =154)	Refusal	133	86.4
	Advised against by the G.P.	16	10.4

* None received 1 dose of vaccine.

** *P*<0.001.

*** P<00.1.

**** None died.

Table 3

Prevalence of anti COVID-19 vaccination coverage in 1512 subjects with liver cirrhosis. : Italy, 2022.

Characteristic		N.vaccinated	N. subjects	%	p value Chi-square test
Sex	Male	944	1052	89.7	>0.05
	Female	414	460	90.0	
Age	< 65 years	654	760	86.1	< 0.001
	\geq 65 years	702	750	93.6	
Area of birth	Italy	1233	1356	90.9	< 0.001
	Abroad	124	155	80.0	
Area of residence	North/Center	752	810	92.8	< 0.001
	South/Islands	606	702	86.3	
YEARS OF SCHOOLING	≤8 years	609	703	86.6	< 0.001
	> 8 years	745	803	92.8	
Etiology*	Not alcoholic	1126	1237	91.0	< 0.05
	Alcoholic	112	132	84.9	
Child	А	1150	1264	91.0	< 0.05
	B/C	207	247	83.8	

* 143 subjects with mixed etiology were excluded.

Table 4

Frequencies (%) of characteristics in 154 not vaccinated subjects by area of birth, Italy 2022.

Characteristic		Italians ($N = 123$)		Foreigners $(N = 31)$		p value Chi-square	
		N.	%	N.	%	Test	
Sex	Male	91	74.0	17	54.8	<0.05	
	Female	32	26.0	14	45.2		
Age	< 65 years	82	66.7	24	77.4	>0.05	
	\geq 65 years	41	33.3	7	22.6		
ETIOLOGY*	Not alcoholic	88	85.4	23	82.1	>0.05	
	Alcoholic	15	14.6	5	17.9		
Child	А	92	74.8	22	71.0	>0.05	
	B/C	31	25.2	9	29.0		
Reasons of not	Refusal	105	85.4	28	90.3	>0.05	
VACCINATION	Other	19	14.6	3	9.7		

* 23 subjects with mixed etiology were excluded.

Table 5

Crude and adjusted Odds Ratios (O.R.) derived by multiple logistic regression analysis for the association of different variables with lack of anti COVID-19 vaccination in cirrhotics. Italy, 2022.

Factor Age	\geq 65 years	Crude O.R. 1.00	95% CI	Adjusted O.R.* 1.00	95% CI
	< 65 years	2.37	1.66 - 3.39	2.88	1.91 - 4.36
Area of birth	Italy	1.00		1.00	
	Abroad	2.51	1.62 - 3.87	2.31	1.43 - 3.75
AREA OF RESIDENCE	North/Center	1.00		1.00	
	South/Islands	2.05	1.46 - 2.89	1.33	0.86 - 2.06
Years of schooling	>8 years	1.00		1.00	
	\leq 8 years	1.98	1.41 - 2.80	1.91	1.25 - 2.92
Etiology**	Not alcoholic	1.00		1.00	
	Alcoholic	1.81	1.08 - 3.03	1.08	0.61 - 1.90
Child	А	1.00		1.00	
	B/C	1.95	1.32 - 2.88	1.59	0.99 - 2.56

* Adjusted for the confounding effect of all listed variables.

** 143 subjects with mixed etiology were excluded.

3.4. Independent predictors of breakthrough infection

The 1137 subjects who had received 3 or more doses of vaccine were compared by occurrence of breakthrough infection. Baseline liver disease stage (i.e. Child class B/C) was the only independent predictor of this outcome (adjusted O.R. 2.30; Cl 95% = 1.37-3.86). No association was found with sex, age, and etiology. (Table 6)

4. Discussion

In Italy, immunization with COVID-19 vaccine was strongly recommended for patients with chronic diseases, including those with cirrhosis, from March 2021. Thereafter, a third and a fourth dose of vaccine were recommended for these subjects since August 2021 and March 2022, respectively. Consequently, subjects enrolled in the study period (January-June 2022) may have received 3 or 4 doses of vaccine.

This is the first study assessing vaccine coverage against COVID-19 among cirrhotic subjects in a European country. The nearly 90% overall vaccine coverage is strong evidence for a very successful vaccination campaign against COVID-19 infection. Even subjects belonging to groups at lower vaccine coverage reported uptake vaccination rates from 80 to 90%. Of particular interest is the 80% rate among people born abroad, which reflects how regular immigrated have the same chance of access to our health care system as the native people in Italy do.

Table 6

Frequencies (%) of characteristics of subjects with \geq 3 doses of COVID vaccine with or without breakthrough infection. Adjusted Odds
Ratios (O.R.) derived by multiple logistic regression analysis. Italy, 2022.

Factor		With Breakthrough $(N = 103)$		Without Breakthrough ($N = 1034$)		Adjusted	
	N.	%	N.	%	O.R.*	95% CI	
Sex	Female	26	25.2	313	30.3	1.00	
	Male	77	74.8	721	69.7	1.25	0.76 - 2.05
Age	\geq 65 years	46	44.7	482	46.7	1.00	
	< 65 years	57	55.3	551	53.3	0.83	0.53 - 1.29
Etiology**	Not alcoholic	82	87.2	885	92.8	1.00	
	Alcoholic	12	12.8	69	7.2	1.56	0.80 - 3.07
CHILD	А	76	73.8	905	87.5	1.00	
	B/C	27	26.2	129	12.5	2.30	1.37 - 3.86

* Adjusted for the confounding effect of all listed variables.

** 89 subjects with mixed etiology were excluded.

The present figures largely exceed those (all from retrospective surveys) of 60% observed in USA [15] among cirrhosis of any stage, and 37.1% in China [16] among decompensated cirrhosis.

Identification of risk factors associated with unvaccinated status is critical to target interventions and further improve vaccine coverage. As in the U.S. study [15], subjects younger than 65 years of age were nearly 3-fold less likely vaccinated, reflecting an attitude of younger people to underestimate the risk of severe infection and thus defer vaccination. Even subjects with lower years of schooling are at increased risk (adj. O.R. 1.91; CI 95% = 1.25-2.92) of lack of vaccination, probably because of vaccine hesitancy stemming from personal or political beliefs. The finding that 86.4% of cases reported refusal as reason for not undergoing vaccination, further supports the previous consideration. Moreover, subjects with lower educational level may be the most at risk of falling prey to vaccine misinformation. Of notice is the observation that nearly 10% of subjects remained unvaccinated since they were not provided correct medical advice by their G.P.s, who probably did not consider vaccines against COVID-19 effective and safe for cirrhotic subjects. Thus, we suggest that counselling on the risks and benefits of COVID-19 vaccination should be specifically addressed to both cirrhotic patients and G.P.s to correct their attitude against vaccination [20]. Correct counselling will also counter the vaccine misinformation provided by some media outlets and online social media [20].

Though breakthrough infections may occur after completion of the primary vaccination cycle, these are associated with a reduced COVID-19 related hospitalization and mortality [11,12]. Our findings further strengthen the favourable effect of vaccination on severe outcomes even in cirrhotic subjects. Indeed, among the 178 subjects with breakthrough infection, only 12 (6.7%) required hospitalization and none died. Moreover, the observed better effectiveness of a third vaccine dose in preventing breakthrough infection and hospitalization, confirms the results of a recent U.S. study [21]. This latter study supports the importance of a third dose of mRNA vaccine among patients with cirrhosis, suggesting that it can also overcome their vaccine hyperresponsiveness [21].

Decompensated cirrhosis (i.e. advanced Child stage) was the only factor independently associated (adj. O.R. 2.30; Cl 95% 1.37–3.86) with breakthrough infection, suggesting that liver status is the major determinant to acquire the infection in vaccinated subjects.

5. Limitations

We acknowledge some potential limitations to our study. The true proportion of breakthrough infections may be underestimated because asymptomatic infections, which more likely occur in vaccinated people [22], weren't identified. Moreover, subjects vaccinated may be less likely to receive COVID-19 PCR testing in the presence of symptoms, and thus their positive status might go undetected. However, these points may have affected the incidence rate of breakthrough infections, but not that of severe outcomes such as hospitalization and death.

6. Strength

The study is characterized by several strong points which are worth underlying. First, the observational study design, with the enrolment of all consecutive patients, generates a more accurate ascertainment than retrospective studies, generally affected by selection and ascertainment biases. Second, the large cohort (1512 cases) enrolled allows an accurate estimate of Odds Ratios for the associations explored. Third, participating centers were scattered all over the country, assuring representativeness to the observed findings. Finally, the enrolment period lasting 6 months corresponds to the interval time recommended by the Italian Association for the Study of Liver (AISF) for surveillance of cirrhotic subjects; consequently, nearly all cirrhotic subjects referring to the participating centres had the chance to be observed.

7. Conclusions

Our findings evidence very high vaccine coverage against COVID infection in Italy among patients with liver cirrhosis. Educational campaigns targeting subjects younger than 65 years of age and with lower years of schooling may further improve vaccination coverage. Patients with decompensated cirrhosis are at higher risk of breakthrough infection. Vaccine is effective in protecting against hospitalization and death. Three doses of vaccine are more effective than two in preventing breakthrough infection and hospitalization.

Conflict of interest

All authors have no conflict of interests.

References

- Marjot T, Moon AM, Cook Ja, et al. Outcomes following SARS-CoV-2 infection in patients with chronic liver disease: an international registry study. J Hepatol 2021;74:567–77.
- [2] Ge J, Pletcher MJ, Lai JC, et al. Outcome of SARS-CoV-2 infection in patients with chronic liver disease and cirrhosis: a national COVID cohort collaborative study. Gastroenterology 2021;61:1487–501.
- [3] Albillos A, Lario M, Alvarez-Moh M. Cirrhosis associated immune dysfunction: distinctive features and clinical relevance. J Hepatol 2014;61:1385–96.
- [4] Bonnel AR, Bunchorntavakul C, Reddy KR. Immune dysfunction and infections in patients with cirrhosis. Clin Gastroenterol Hepatol 2011;9:727–38.
- [5] Polack FP, Thomas SJ, Kitchin N, et al., C4591001 Clinical Trial Group Safety and efficacy of the BNT162b2 mRNA COVID-19 vaccine. N Engl J Med 2020;383:2603–15.
- [6] Baden LR, El Sahly HM, Essink B, et al., COVE Study Group Efficacy and Safety of the mRNA-1273 SARS-CoV-2 Vaccine. N Engl J Med 2021;384:403–16.

- [7] Mccashland TM, Preheim LC, Gentry MJ. Pneumococcal vaccine response in cirrhosis and livertransplantation. | Infect Dis 2000;181:757–60.
- [8] Harmala S, Parisinos CA, Shallcross L, et al. Effectiveness of influenza vaccines in adults with chronic liver disease: a systematic review and meta-analysis. BMJ Open 2019;9:031070.
- [9] Aggeletopoulou I, Davoulou P, Konstantakis C, et al. Response to hepatitis B vaccination in patients with liver cirrhosis. Rev Med Virol 2017;27:e1942.
- [10] Thuluvath PJ, Robarts P, Chauhan M. Analysis of antibody responses after COVID-19 vaccination in liver transplant recipients and those with chronic liver diseases. J Hepatol 2021;75:1434–9.
- diseases. J Hepatol 2021;75:1434-9.
 [11] John BV, Deng Y, Scheinberg A, et al. Association of BNT162b2 mRNA and mR-NA-1273 vaccines with COVID-19 infection and hospitalization among patients with cirrhosis. JAMA Intern Med 2021;181:1306-14.
- [12] John BV, Deng Y, Schwartz KB, et al. Postvaccination COVID-19 infection is associated with reduced mortality in patients with cirrhosis. Hepatology 2022;76:126–38.
- [13] Fix OK, Blumberg EA, Chang KM, et al. AASLD expert panel consensus statement: vaccines to prevent coronavirus disease 2019 infection in patients with liver disease. Hepatology 2021;74:1049–64.
- [14] Cornberg M, Buti M, Eberhardt CS, et al. EASL position paper on the use of COVID-19 vaccines in patients with chronic liver diseases, hepatobiliary cancer and liver transplant recipients. J Hepatol 2021;74:944–51.
- [15] Mahmud N, Chapin SE, Kaplan D, et al. Identifying patients at highest risk of remaining unvaccinated against severe acute respiratory syndrome coro-

navirus 2 in a large veterans health administration cohort. Liver Transplant 2021;27:1665–8.

- [16] Cao Z, Zhang C, Zhao S, et al. COVID-19 vaccines in patients with decompensated cirrhosis: a retrospective cohort on safety data and risk factors associated with unvaccinated status. Infect Dis Poverty 2022;11:56–64.
- [17] Gaiani S, Gramantieri L, Venturoli N, et al. What is the criterion for differentiating chronic hepatitis from compensated cirrhosis? A prospective study comparing ultrasonography and percutaneous liver biopsy. J Hepatol 1997;27:979–85.
- [18] Chon YE, Choi EH, Song KJ, et al. Performance of transient elastography for the staging of liver fibrosis in patients with chronic hepatitis B: a meta-analysis. PLoS One 2012;7:e44930.
- [19] Stroffolini T, Sagnelli E, Gaeta GB, et al. Characteristics of liver cirrhosis in Italy: evidence for a decreasing role of HCV aetiology. Eur J Intern Med 2017;38:68–72.
- [20] Abbasi J. COVID-19 conspiracies and beyond: how physicians can deal with patients' misinformation. JAMA 2021;325:208-10.
- [21] John VB, Ferreira RD, Doshi A, et al. Third dose of COVID –19 mRNA vaccine appears to overcome vaccine hyporesponsiveness in patients with cirrhosis. J Hepatol 2022;77:1349–58.
- [22] Aschwanden C. Five reasons why COVID herd immunity is probably impossible. Nature 2021;591:520–2.